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Discussion: 

NPOL and D3R are green. XPOLs 2-4 are green. XPOL 5 is yellow but will be placed back into
operat ion soon. Disdrometers and other surface instrumentat ion are most ly green. SN25 was
re-calibrated yesterday, but calibrat ions for other sites were found to be stable. TheAPU07
computer was damaged by a f lash f lood event and so the instrument is red for the remainderof
the campaign.

Lit t le precip occurred in the IFloodS domain yesterday. Early this morning, a most ly
strat iformrain systemwith some leading convect ion has moved thru our domain and has
overrun the NPOL/D3R site. Coordinated scanning cont inues on this event.


